Author's response to reviews

Title: Randomised controlled trial of a supervised exercise rehabilitation program for colorectal cancer survivors immediately after chemotherapy: study protocol

Authors:
Rosalind R Spence (rspence@hms.uq.edu.au)
Kristiann C Heesch (kheesch@hms.uq.edu.au)
Elizabeth G Eakin (e.eakin@sph.uq.edu.au)
Wendy J Brown (wbrown@hms.uq.edu.au)

Version: 2 Date: 25 July 2007

Author's response to reviews:

Dear Editor,

Thank you for confirming the acceptance of our manuscript for publication in BMC Cancer. As requested, the following formatting changes have been made:

- Methods/design - this section was renamed 'Methods/Design' in both the abstract and body of the manuscript.
- Table title and legend - The table title now appears above the table. The legend now appears below the table.
- Additionally a number of typing errors were identified and fixed by the authors. We believe it is now ready for publication.

Kind Regards,

Rosa Spence